Author’s response to reviews

Title: Cysticlean® a highly PAC standardized content in the prevention of recurrent urinary tract infections: an observational, prospective cohort study.

Authors:

Francisco Sanchez Ballester (sanchezballester@gmail.com)
Vicente Ruiz Vidal (vte.ruiz.vidal@gmail.com)
Emilio Lopez Alcina (eloalc@yahoo.es)
Cristina Domenech Perez (cristina.domenech@uv.es)
Eva Escudero Fontano (e.escudero.fontano@gmail.com)
Juan Yuste Garrido (jyuste75@gmail.com)
Ana Montoliu Garcia (ana_montoliu@hotmail.com)
Ana Oltra Benavent (anamari.oltra@gmail.com)

Version: 3 Date: 21 May 2013

Author’s response to reviews:

Dear Associate Editor,

Attached you can find answers to your comments about the manuscript “Cysticlean® a highly PAC standardized content in the prevention of recurrent urinary tract infections: an observational, prospective cohort study”.

We have made appropriate remarks in the manuscript.

Thank you in advance and best regards,

Dr. Sánchez

1) Reviewer Associate Editor

“The first paragraph of discussion should be placed as the last one.”

Done.

“The authors must emphasize the role of a randomized placebo controlled study in the last paragraph.”

Done.